Author's response to reviews

Title: The disease burden of pertussis in adults 50 years old and older in the United States: a retrospective study

Authors:

Catherine Balderston McGuiness (kmcguiness@us.imshealth.com)
Jerrold W Hill (j.hill@us.imshealth.com)
Eileen M Fonseca (efonseca@us.imshealth.com)
Gregory P Hess (ghess@us.imshealth.com)
William P Hitchcock (lajollapedspeds@yahoo.com)
Girishanthy X Krishnarajah (girishanthy.x.krishnarajah@gsk.com)

Version: 4 Date: 30 November 2012

Author's response to reviews: see over